Author's response to reviews

Title: The Scandinavian Multicenter Hemodynamic Evaluation of the SJM Regent Aortic Valve

Authors:

Jon Offstad (jon.offstad@rikshospitalet.no)
Kai Andersen (Kai.Andersen@Rikshospitalet.no)
Per Paulsson (per.paulsson@skane.se)
Jesper Andreasson (jesper.andreasson@skane.se)
Ulf W Kjellman (ulf.w.kjellman@vregion.se)
Oluf Lundblad (olof.lundblad@orebro.se)
Gunnar Engstøm (gunnar.engstrom@med.lu.se)
Rune Haaverstad (rhaavers@online.no)
Jan L Svennevig (j.l.svennevig@medisin.uio.no)

Version: 2 Date: 30 October 2011

Author's response to reviews:

We agree with the reviewers comments.

1. A description of the clinical material has been included under Material and Methods
2. The Dobutamine stress test was performed whenever available
3. Clinical results have been included under Results
4. The study received a grant. This information has been listed under Acknowledgements.
5. Hemodynamic data at rest is listed in the tables.